Author's response to reviews

Title: Clinical presentation and diffusion weighted MRI of acute cerebral infarction. The Bergen Stroke Study.

Authors:

Halvor Naess (haln@haukeland.no)
Jan C Brogger Jr (jcbr@helse-bergen.no)
Titto Idicula (idic@helse-bergen.no)
Ulrike Waje-Andreassen (uwan@helse-bergen.no)
Gunnar Moen (gmoe@helse-bergen.no)
Lars Thomassen (ltho@helse-bergen.no)

Version: 2 Date: 16 May 2009

Author's response to reviews:

Statements as to approval by the local ethical committee and informed consent is added to the Methods.
Median time from stroke onset to CT and MRI is added to the Results as suggested by referee 2.
Some abbreviations have been deleted in the Discussion as suggested by referee 2.
Competing interests and author’s contributions have been added.
Reference style has been changed conform with the journal style.